Is Wealth Causing Health? Evidence from Stock Market Induced Wealth Shocks.

Hannes Schwandt^{*} Universitat Pompeu Fabra

Draft, March 2011

Abstract

This paper investigates the effect of wealth shocks on retiree health, mental health and mortality. Stock market fluctuations over the past 18 years are exploited as a natural experiment that generated considerable gains and losses in the wealth of US retirees. Using data from the Health and Retirement Study I find that a 10% wealth increase leads to an improvement of 1.5-3% of a standard deviation in each of four broad physical and mental health measures including mortality. Effects are heterogeneous across different physical health problems, with most pronounced effects for the incidence of high blood pressure, smaller effects for heart problems, strokes and cancer, and no effects for arthritis, diabetes and lung diseases. These results suggest that wealth shocks strongly affect retiree health in the US and that psychological factors seem to be an important channel through which these effects are delivered.

JEL Classification: I10; J14; J26 Keywords: health; mortality; wealth shocks; stock market; US retirees

^{*}Email: hannes.schwandt@upf.edu; I am grateful to Steve Pischke for long discussions and very helpful supervision. I am also thankful to Antonio Ciccone, Dora Costa, Janet Currie, Robin Hogarth, Micheal Hurd, Richard Layard, Adriana Lleras-Muney, Susann Rohwedder, James Smith, Till von Wachter and seminar participants at LSE, UPF, EDP Jamboree Bonn, RAND and UCLA for helpful comments.

1 Introduction

It is a well-established fact that richer people are healthier, happier and live longer. However, it is little understood to which extent this relationship is driven by a causal effect of wealth on physical health, mental health and mortality. Money might buy health, but health also might reversely affect income and expenditure. And third factors, such as education or ability, are likely to influence both wealth and health simultaneously. The existing literature on the wealth-health relationship finds little evidence for a causal effect of wealth on adult health in developed countries and tends to emphasize reverse causality running from health to income.¹ Contrary to these findings, this paper provides evidence that wealth shocks have a strong positive causal effect on physical health, mental health and mortality of wealthy retirees in the US.

To identify the causal effect of wealth on adult health, this paper exploits stock market induced shocks in the wealth of US retirees. Over the past two decades about one-third of US retiree households held part of their wealth in stocks. And for stock-holders the fraction of life-time wealth held in stocks amounted on average to about 20%. As a consequence the booms and busts in the US stock market over the past two decades generated considerable unexpected gains and losses in the wealth of stock holding retirees. Using data from the Health and Retirement Study (HRS) I predict wealth shocks by the interaction of individual stock holdings with stock market changes and analyse their effect on various health measures. I find that a 10% change in life-time wealth over a two year period is associated with a change of 1.5-3% of a standard deviation in four different health measures: A health problems index, self-reported health, mental health and the probability to survive two more years. Analysing individual health problems indicates a strong effect on hypertension, smaller effects on heart diseases, strokes and cancer, and no effect on arthritis, diabetes and lung disease. These effects are surprisingly large. The comparison with the overall wealth-health relationship in the sample suggests that estimated effects are of equal size or even larger then the overall relationship.

Wealth shocks are predicted by the interaction of the fraction of wealth held in stocks in the past period with the stock market change between the past and the current period. To interpret their association with changes in health as causal effects of wealth on health, predicted wealth shocks must satisfy two condition. They must be independent of any unobserved heterogeneity and their effect on health must exclusively run through changes in stock wealth.

As stock holdings are likely to be endogenous I control separately for the fraction of wealth held in stocks. In other words, I compare health changes for individuals with the same fraction of wealth held in stocks at different points in the stock market cycle. One concern might be that a retiree with 20% of her wealth in stocks right before a boom might not be

¹For reviews of the literature see Smith (1999), Deaton (2003), Cutler, Deaton and Lleras-Muney (2006), Cutler, Lleras-Muney and Vogl (2010).

comparable to a retiree with 20% in stocks right before a bust. To rule out that estimates are driven by such correlation of unobservables with the stock market cycle I present regressions in which the individual stock fraction is instrumented by the individual's stock fraction in the first wave or by the average stock fraction over the whole sample. Both initial and average stock fractions are independent of were we are in the stock market cycle.

The exclusion restriction, that predicted shocks affect health exclusively through the change in stock wealth, might be invalidated for several reasons. Stock market changes could be correlated with changes in other wealth components such as house prices. Such correlation would bias the wealth shock coefficient because predicted wealth shocks would systematically under- or overestimate actual wealth shocks. Looking at changes in reported wealth for retirees with and without stocks in the previous period, however, indicates that this is unlikely to be the case. While reported wealth of stock holding retirees strongly correlates with the changes in the stock market, the correlation for those without stocks is essentially zero. The exclusion restriction would further be invalidated if the stock market or more generally the macroeconomic environment had non-wealth effects on retiree health. I argue that retirees without stocks are at least equally strongly affected by such direct effects as those with stocks. Therefore I include time effects to absorb any macroeconomic shocks common to both groups.

Several robustness checks give support that estimated effects actually represent causal effects of wealth shocks on health. Further, effects are quite homogeneous across age, gender and negative vs. positive shocks. Only for mortality effects are especially strong for retirees above age 75 and if shocks are negative.

The evidence of positive effects of wealth shocks on health presented in this paper is in contrast with much of the existing literature. Adams et al. (2003), Smith (2005) and Michaud and Van Soest (2008) find that lagged wealth, income and other components of SES do not predict changes in health for elderly in the US. Ruhm (2000) analyses mortality-unemployment state-level time series finding that higher unemployment is associated with lower mortality. Deaton and Paxson (2001) and Deaton and Paxson (2004) look at mortality-income cohort times series for the US and the UK, respectively. Their results indicate that periods of greatest mortality declines do not match with periods of greatest income increases. Snyder and Evans (2006) compare cohort mortality around the so called 'Social Security notch'² finding that the notch generation which received lower Social Security had slightly lower mortality.

It is, however, difficult to derive causal inference about the effects of wealth on health from these studies. For the first group of papers, which employes an approach closely related

 $^{^{2}}$ In the 1970s a calculation error lead to unexpected increases of Social Security benefits for the 1910-1921 birth cohorts. The error was corrected in 1977, abruptly decreasing benefits and generating the so called 'Social Security notch'.

to granger causality, it is not clear whether lagged income or wealth is truly exogenous.³ Also wealth and income at the individual level are notoriously subject to measurement error which makes it -in the absense of an instrument- hard to distinguish a zero effect from attenuation bias. The second branch of papers tackles endogeneity using aggregate income variation that cancels out unobserved heterogeneity at the individual level. But aggregate shocks that generate or correlate with such aggregate income variation are likely to have also non-income effects on health.⁴ This invalidation of the exclusion restriction, as the authors of these papers note, makes it difficult to infer a zero or negative causal effect of wealth on health from these findings.

So far only a few papers identify income shocks at the individual or household level. Jensen and Richter (2003) look at the pension crisis in Russia during which many retirees were not paid their pensions for an extended period of time. They find that for affected pensioners, who faced on average an income decrease by 24%, nutrition and use of health care decreases while the likelihood to die over the next two years increases by 5%. Case (2004) analyses the effect of a pension reform in South Africa that lead to a large and unexpected income increase for black and coloured pensioners. She finds positive health effects, which run through improved nutrition, living conditions and psychological factors such as a reduction of stress. Lindahl (2005) uses lottery winnings in Sweden as source of exogenous income variation estimating that a 10% income increase improves self-reported health by 4-5% of a standard deviation. Sullivan and von Wachter (2008) find that exogenous job displacements increase the mortality hazard of male US workers by 50-100% during the years following the job loss, while long-run earnings decrease by 15-20%. These estimates are similar to the mortality effects found in this study, though they are likely to be driven to some part by non-income effects of unemployment on health.

Should we, however, expect positive effects found for poor retirees in developing countries or displaced workers in the US to carry over to wealthy, stock holding US retirees? Health inputs like medical treatment, medication or mere calorie intake might be affected by wealth shocks for poor retirees in Russia or South Africa. But this is probably not the case for stock holding US pensioners, who have even after a considerable wealth loss enough money left to affort basic pills and burgers. Further medicare covers the entire 65+ population in the US so that wealth shocks do not affect health insurance coverage unlike for displaced workers. Consumption of healthy food and purchase of a healthy environment could be more responsive determinants of retiree health in the US than basic health inputs. But two years might not be enough time for luxury goods to become manifest in physical health outcomes.

 $^{^{3&}quot;}$ [T]here are influences between income and health that run in both directions, and the lags can be as long as a human lifetime." (Deaton 2003, p. 121). For further discussion see Mealli and Rubin (2003), and Cutler (2005).

 $^{^{4}}$ For example, the cause of death most (negatively) affected by the unemployment rate in Ruhm (2000) are care accidents, which is probably due to less traffic in times of less economic activity rather than a better driving style of unemployed people. Similarly, changes in cohort mortality over the time period analysed by Deaton and Paxson (2001) and Deaton and Paxson (2004) were probably dominated by technical innovation and new diseases (e.g. Aids).

A more plausible channel would be psychological factors such as happiness about pleasant trips that were not affordable before or stress and sadness about a lost fortune that was planned as an inheritance for the grand children. Such channel would be in line with much of the literature in medicine, psychology and biology, that emphasizes psychological factors as important determinants of mental and physical health. There is broad agreement that stress causes depression and coronary artery diseases (for a review see Strike and Steptoe 2004), which in turn increase the likelihood of other severe diseases such as heart diseases and strokes and eventually death. Moreover various studies have suspected stress to be a driver for certain cancer types (for a review see Chida et al. 2008). There is also some agreement that positive emotions have a positive effect on health (for a review see Chida and Steptoe 2008).⁵ This paper finds strong wealth shock effects on high blood pressure and mental health and smaller effects on heart problems, strokes and cancer. Given the circumstances and the timing of the wealth shocks as well as the findings from the bio-medical literature, this seems to be a plausible effect pattern.

The focus of this study on retirees is both an advantage and a limitation. Compared to younger adults retirees have a lot of wealth and heterogeneity in wealth composition so there is a lot of variation to exploit. And as they do no longer participate in the labour market effect of stock market shocks running through labor demand as well as labor supply responses are limited. That makes it easier to separate wealth shocks from other confounding effects. Further the elderly are closer to the margin of severe health conditions than younger adults so that wealth effects on health are more likely to become manifest in measurable health outcomes. On the downside, estimates of wealth shock effects on retiree health might not be comparable to effects of wealth on health for younger adults that participate in the labor market. The latter are more flexible and in better physical shape and therefore likely to be less affected by wealth shocks. The results of Sullivan and von Wachter (2008) indicate, however, that this is not necessarily the case.

Section II describes the data set I use, while Section III discusses the identification strategy in detail that leads to the empirical specification in Section IV. Section V presents the findings and Section VI concludes.

⁵For example, in experiments healthy subjects were exposed to rhinovirus or influenza virus and those who scored ex-ante higher on optimism and happiness scales were less likely to get infected (Cohen et al. 2006).

2 Data

The data used in this study comes from the Health and Retirement Survey (HRS). The HRS is a biannual panel starting in 1992 with 12,654 individuals representing US adults aged 51 - 61 in 1992 (born during the years 1931 - 1941). In 1998 and 2004 new cohorts were added to keep the sample representative for those of age 51 and older. This study uses 9 HRS waves (1992 - 2008) covering over 170,000 person-year observations of about 30,000 individuals from 20,000 households. The analysed sample is restricted to singles and couples that were retired in the previous wave (i.e. both respondent and spouse - if existent- were neither working nor unemployed) and provided information on wealth, stock holdings and retirement income. The final regressions sample uses about 90,000 person-year observations. Of these, 40,000 refer to a single response per household-year (no spouse existent) while the remaining are responses of two spouses (two observations per household-year). The interview month is known, so that the HRS data can be matched to monthly stock market data from the Standard & Poor's 500 (S&P500).⁶

a. Wealth data

The HRS contains detailed information on income and wealth holdings.⁷ Wealth and individual income is reported by one 'financial respondent' per household and matched to the spouse if existent. Therefore wealth varies only between households while health measures and other individual variables differ between spouses. Financial information is reported in exact amounts and unfolding response brackets are offered if exact amounts are unknown.

a.1 Current household wealth

Household wealth consists of: net housing wealth⁸; real estate wealth; vehicles; business wealth; individual retirement accounts (IRAs); stocks and mutual funds; checking and savings accounts; cds, savings bonds and treasury bills; bonds; other savings; debts.

a.2 Life-time wealth

Life-time wealth (W_t) is the discounted sum of current wealth and future income.

$$W_t = A_t + \sum_{\tau=0}^{T-t} Y_{t+\tau} \frac{S_{t,\tau}}{(1+r)^{t+\tau}}$$

 A_t is current wealth, $Y_{t+\tau}$ future earnings, r the real annual interest rate and $S_{t,\tau}$ the individual survival rate. Current wealth and *past* earnings are well documented in the HRS. Fortunately, retiree income - consisting of pensions, annuities, old age social security and veteran benefits - can be expected to stay constant after the first receipt until the

⁶Interviews which start in one month and end in a later month were dropped as well as spouse interview's that were not conducted in the same month as for the financial respondent.

⁷This study uses cleaned and partly imputed wealth data from the 'RAND Income and Wealth Imputations' file. Variables that are not included in the RAND file were added from the HRS raw data.

⁸Housing wealth does not include secondary homes, as this information is not available in wave 3.

individual's end of life (receipts stay constant in real terms). Hence for retirees we can simply take past year's annual income from pensions, annuities, old age social security and veteran benefits as the expectation for future income.⁹ Interest rate expectations (current values taken from the SSA report) are assumed to stay constant as well. (τ)-year survival rates are constructed by gender (g), 10-year birth cohort (c) and age (t) using the SSA life tables.¹⁰

$$W_{t} = A_{t} + (SS_{t} + PAI_{t} + VetBen_{t})\sum_{\tau=1}^{T-t} \frac{S_{t,\tau,g,b}}{(1+r)^{t+\tau}}$$

Social security benefits pose a potential problem as their receipt does not depend on labor force status and even though people can claim social security at age 62 (if entitled) there are financial incentives to postpone social security take-up to age 65 (Coile et al. 2002). Thus for retirees below age 65 who do not report to receive social security it is not clear whether they are postponing or whether they are not entitled to social security payments and life-time wealth cannot be constructed. Fortunately the HRS includes a question about future expected social security payments, so that it can be tested whether results are sensitive to the exclusion of retirees who do not yet get social security but expect payments in the future.

Another complication are different life expectancies within households, i.e. within couples. Typically wives can expect to survive their husbands, but it would be demanding to calculate all different survival constellations and the corresponding exact survivor benefit amounts. For simplicity a couple's life-time wealth is calculated by applying the couple's mean life expectancy to the sum of the couple's total annual income.

$$W_{t} = A_{t} + (SS_{t} + PAI_{t} + VetBen_{t} + SS_{t}^{s} + PAI_{t}^{s} + VetBen_{t}^{s})\sum_{\tau=1}^{T-t} \frac{S_{t,\tau,g,b} + S_{t,\tau,g,b}^{s}}{2(1+\tau)^{t+\tau}}$$

Restricting the sample to singles in order to avoid this simplified life-time wealth formula does not affect the pattern of the estimated effects, only standard errors increase due to the decreased sample size.

a.3 Stock holdings

A central ingredient of the measure of wealth shocks used in this study is the amount of stock holdings. Direct stock holdings are well documented in each wave, but they do not include stocks held in IRAs. Retirees often hold considerable fractions of their wealth in (often various) IRAs. To calculate the total amount of stock holdings it is therefore

 $^{^{9}\}mathrm{The}$ HRS usually reports monthly (past month's) income which is multiplied by 12 to obtain future annual income.

¹⁰One might worry about the direct influence of age, gender and birth cohort trough the survival rate on predicted wealth shocks, which are rescaled by life-time wealth. However, results are robust against different specifications of the survival rate. Even setting it to equality (taking only next year's income instead the whole life-time sum) does not change results qualitatively.

important to known the percentage of each IRA invested in stocks.

In 2006 and 2008 for each IRA the exact percentage invested in 'stocks and mutual funds' (i.e. not in interest-earning assets) is reported. In 1998 - 2004 three categories indicate whether IRAs are invested 'mostly in stocks', 'mostly in interest-earning assets', or 'about evenly split'. I translate these categories into 100%, 0%, and 50% invested in stocks, which results in roughly the same investment distribution in 2004 as for the exact information in 2006 and 2008. The assumption of a stable investment distribution between 2004 and 2006/2008 for US IRAs is checked with data from the Survey of Consumer Finances (SCF), a US representative triennial survey with about 22,000 households per wave. The SCF reports exact information on the IRA fraction invested in stock for 2004 and 2007. The cumulative distribution function does not change significantly between SCF 2004 and SCF 2007, indicating that IRA investment distributions in the US were indeed stable over that period.

For the three initial HRS waves, 1992 - 1996, no information is available on IRAs invested in stock. In order not to lose these entire waves, IRAs in these years are assigned the average IRA stock investment rate of the year 1998 (52%). This adds a considerable amount of noise and results should be tested against the exclusion of these waves. Notice that replacing a regressor by its mean does not lead to 'classical' measurement error, i.e. error uncorrelated with the true value and correlated with the included regressor. Instead the resulting error is correlated with the true value but not with the regressor.¹¹ Therefore estimates are unbiased, only standard errors are increased.

a.4 Wealth summary statistics

Table 1 summarizes the number of observations, age, retiree rates and main wealth measures. In 1998 older than average cohorts are added and younger cohorts in 2004, leading to discontinuous jumps in these measures. Retiree rates increase with age, but even at age 70 for 30% of the households at least one spouse is still in the labor force. The fourth and fifth row show the information available on the fraction of IRAs invested in stocks and the respective imputed values. The regression sample includes all households that in the *previous* wave (i) were retired and (ii) reported wealth, retiree income (for the construction of life-time wealth) and stock holdings. Hence there are no observations in the first wave and the wealth information in wave 1 refers to the number of observations in wave 2. In the regression sample on average about half the life-time wealth is held in current wealth and about 1/3 of all households hold at least some stocks. Since wealth shocks are predicted for households with stocks, these are the 'treated'. They are on average twice as wealthy as retirees without stocks and hold about 20% of their life-time wealth in stocks.¹² For a

¹¹Think of the error as the residual from the regression of the true value on a constant; the constant is the mean and orthogonal to the residual

 $^{^{12}}$ In the first three waves the fraction of households with stocks is inflated due to the assumption that any IRA is invested 52% in stocks. For the same reason stock holdings in these waves are artificially low because many poor households with small IRA accounts that in reality do not own any stocks are included in the group of stock holders.

discussion of wealth changes of these different groups see the next section (Figure 4 and Tables 3, 4).

b. Stock market data

To predict the average gains and losses in stock portfolios, stock holdings are interacted with changes in the Standard and Poor's 500 (S&P500) index. The S&P500 is the weighted average of 500 of the biggest actively traded companies in the US and therefore represent a broad indicator of the US stock market.¹³ Figure 1 shows the S&P500 index and periods in which HRS interviews were conducted marked by boxes. As it can be seen, the HRS interviews cover well the two major boom and bust periods the US economy experienced over the past 28 years.

Interview months are known so that the HRS data are not matched to only one S&P500 value per wave but on a monthly basis, resulting in 108 instead just 9 distinct matched S&P500 values. Even more stock market variation results for individual first differences. Since individuals are interviewed in different months each wave, the change in stock market prices depends on both the month of the current as well as the month of the past interview. The roughly 45,000 individual first differences in the regression sample are matched to about 900 distinct first differences in stock prices. The distribution of these first differences for the S&P500 is shown in Figure 2. However, HRS respondents probably do not exactly know the current value of the S&P500. If we assume that respondents know the value rounded by 50, then we are left with only 36 distinct S&P500 changes in first differences (about the number of different bars in Figure 2).

Figure 3 plots predicted life-time wealth shocks and the S&P500 over time. Wealth shocks roughly range from -40% to +40%. These are dramatic changes. For a retiree who has about 10 years remaining to live a 10% loss in life-time wealth already equals the amount of planned expenditures for a whole year. If she is smoothing consumption, she will have to spend 10% less than planned every month until the end of her life. If a fixed part of her wealth is planned for inheritance, consumption has to decrease by even more. Notice that predicted wealth changes and the S&P500 comove by construction, therefore the comovement in Figure 3 does not tell us anything about the accuracy of the predictions, i.e. whether the stock market actually affects wealth of stock owning retirees (for a comparison with reported wealth changes see below).

c. Health measures

As dependent variable this study uses different health measures from the HRS: An index of health problems, self-reported health status, a mental health index and mortality. For better comparability of the different measures I follow an approach of van Praag and Ferrer-i-Carbonell (2008) and assign to the categories of each measure the expected value of a standard normal variable conditional on being between the category's lower and upper

 $^{^{13}{\}rm The}$ Dow Jones Industrial Average, for example, includes only 30 companies. However, using the Dow Jones instead of the S&P500 delivers similar results.

cut-off points implied by an ordered probit fitted on the raw sample fraction (van Praag and Ferrer-i-Carbonell refer to this as 'probit-adapted OLS'). Standardising the measures by subtracting the mean and dividing by the standard deviation leads to very similar results.¹⁴

c.1 Index of health problems

The health problem index equals 8 minus the sum of indicators for whether a doctor has *ever* told the respondent that she had a particular disease. The eight included diseases are high blood pressure, diabetes, cancer, lung disease, heart disease, stroke, psychiatric problems, and arthritis. An 8 refers to a respondent who never had any of these diseases diagnosed while a 0 indicates that every disease had been diagnosed at some point. Respondents who report the diagnosis of a health problem in the past but not in the current wave are excluded. First differences are equal or smaller zero.

c.2 Self-reported health and self-reported changes in health

For self-reported health respondent are asked to rate their current health as poor, fair, good, very good or excellent. An additional question, self-reported changes in health, asks whether compared to the previous interview health is worse, the same, or better.¹⁵

c.3 Mental health index

As a mental health index I use the inverse of the CESD depression score: 6 plus 2 'positive' indicators minus 6 'negative' indicators (i.e. 0=worse, 8=best). The negative indicators measure whether the respondent experienced the following emotions all or most of the time during the past week: felt depressed, everything is an effort, sleep is restless, felt alone, felt sad, and could not get going. The two positive indicators measure whether the respondent felt happy and enjoyed life, all or most of the time during the past week.

c.4 Mortality

Deaths of survey participants are documented since the third wave (1996). In so-called exit surveys a proxy respondent (usually a surviving family member) is interviewed about time and circumstances of the death. Thus deaths are well documented and not just one possible reason for an observed panel attrition. Table 2 lists the number of documented deaths in the HRS and the regression sample over time. In 1994 -1995 additional and on average much older cohorts (the AHEAD sample) were added, explaining the increase in documented deaths between 1995 and 1996. For 2008 so far only a fraction of all deaths have been documented. 'Survival', used as dependent variable in the baseline regressions, indicates whether the respondent survives until the next interview and it is regressed on past wealth shocks. Hence the mortality regression sample includes those who were retired and reported their wealth (incl. stock holdings) at least two periods prior to their death.

¹⁴The advantage of probit-adapted OLS over standardising is that it takes into account a possibly unbalanced distribution of the sample over the measure's different categories.

¹⁵Wave 1-6 offers 5 categories to rate the health changes: much worse, somewhat worse, same, somewhat better, much better. For comparability with wave 7-9 the first two and the last two categories are recoded as worse and better, respectively.

3 Identification

This paper seeks to estimate the causal effect of wealth on health. The challenge of this task is the endogeneity of wealth. Wealth does not only affect health but in turn is as well affected by health, and third factors might influence both simultaneously. The ideal experiment to solve the problem of endogeneity and to measure the causal effect of wealth on health would be a lottery that randomly assigns wealth losses and gains to people and measures their health before and some time after the assignment.

This paper exploits the booms and busts of the US stock market over the past two decades as a natural experiment that generated considerable wealth gains and losses for those owning stocks. For a causal interpretation it is required that these stock market induced wealth shocks are conditionally independent ('as good as randomly assigned') and that they affect health only through the change in stock wealth ('exclusion restriction').

The data come from the Health and Retirement Survey (HRS) which interviews individuals every second year in detail about their wealth and different measures of health. The interview date is known so that observations can be matched with stock market data on a monthly basis. Given that interview dates are independent of the stock market (or of respondent characteristics, or both), stock market changes between two interviews of the same individual are random.¹⁶ But stock market induced wealth changes are determined by the interaction of stock market changes with stock holdings and stock holdings are probably not randomly assigned. Therefore it is necessary to control for stock holdings separately. This means that I compare retirees with the same fraction of wealth in stocks over different points in time. One might worry, though, that people with the same fraction in stocks before a boom and before a crash are not comparable. A retiree with 20% wealth in stocks at the beginning of a boom might not just be lucky but better informed than a retiree with 20% in stocks right before a crash. Note that such predictability of stock market returns would contradict well-established evidence of market efficiency from the finance literature (for a review see Malkiel 2003). To rule out that results are driven by stock market predictability I show robustness checks in which I instrument stock holdings by average or initial stock holdings, which are both uncorrelated with were we are in the stock market cycle.

The exclusion restriction requires that stock market induced wealth shocks affect health exclusively through stock wealth. First, stock market changes might correlate with house prices or the valuation of other wealth holdings. Due to HRS data restrictions I cannot include regional information on house prices. But I can compare changes in reported wealth

¹⁶Serious health conditions might cause interview delays (Adams et al. 2003, p. 12) which could be a problem if stock market changes are correlated with the length of the time interval. I include the number of months between interviews as a control to take care of this potential endogeneity.

for those with and without stocks. The exclusion restriction would be invalidated if wealth holdings of those without stocks were affected by the stock market. Figure 4 shows that this is not the case, comparing the S&P500 with the coefficients from regressions of wealth changes on time dummies for retirees with stocks (bars) and without stocks (circles) in the previous period. Wealth changes for retirees without stocks are positive in all waves and seem uncorrelated with the stock market. For retirees with stocks they follow the up's and down's in the S&P500, especially after wave 4 when t-1 stock holdings are measured more precisely. Table 3 shows the regression underlying Figure 4 (for regressions using individual changes instead of wave aggregates see below).

Further, stock markets (or more broadly the macroeconomic environment) might affect health also through non-wealth channels. For example, a macroeconomic environment in which stock markets collapse might have negative effects on the individual's employment which would probably not only affect her wealth but also directly her health. To rule out such labour market interactions the sample is restricted to retired households. But there might still be direct effects of stock markets on retiree health, e.g. stress and fears of social instability due to bad news in the media. I include time effects to absorb any macroeconomic shock that is independent of stock holdings and therefore common to retirees with and without stocks. Notice that direct health effects of stock market news might be different for these two retiree groups as those without stocks are on average poorer and probably more risk averse. However, I would expect retirees without stocks to be - if anything - more vulnerable to bad news about the economy which would bias results downwards, making it harder to find significant effects.¹⁷ To show that results do not depend on this assumption I present robustness checks excluding those without stocks.

Finally, predicted wealth shocks could under- or overestimate actual wealth shocks if retirees' expectations of stock market returns systematically divert from zero. The last four HRS waves include stock market expectations which indicate a slight correlation with the stock market. As a consequence net-of-expectations wealth shocks tend to be slightly higher than predicted wealth shocks. However, as expectations are only marginal compared to actual stock market changes (ranging from about -6% to +6% compared to stock market changes of -100% to +100%) their inclusion decreases estimates only slightly (results not yet reported in this draft). The overall effect pattern and the main conclusions remain unchanged, while it makes my results less comparable to other studies that do not include expectations.

The identification strategy outlined above leads to a straight-forward setup: The interaction of stock holdings with stock market changes identifies the causal effect of wealth once I control for the main effects. Holding constant the portfolio composition ensures that wealth shocks are randomly assigned and controlling for time effects absorbs macro shocks that might affect health through other channels besides stock wealth. To decide

¹⁷For example, poorer retirees might be more dependent on public goods like social housing or food programs.

for the best empirical specification a few issues remain to be discussed.

First, the effect of a wealth change on health is likely to depend on the initial wealth level. A \$50,000 loss might not be noteworthy for the very rich but is very painful for somebody poor. And what matters is not just current wealth but life-time wealth, i.e. the discounted sum of current wealth holdings and future income. If you still have many years to live any wealth shock can be smoothed over more periods. And if you can expect to earn a lot in the future a given wealth shock can be better compensated by dissaving. Therefore I divide, or rescale, predicted wealth shocks by life-time wealth to estimate more homogeneous -and more meaningful- average effects over different wealth groups. Note that rescaling just by current instead of life-time wealth would overestimate the relative size of shocks for retirees with high pensioner income and exclude those with zero or negative current wealth.

Second, wealth shocks are predicted by the interaction of past period's fraction of wealth held in stocks with the stock market change since then, ignoring any wealth information from the current period. Why not instead taking reported wealth in both the past and the current period and instrument the change by the stock market changes times stock holdings? Such instrument would be misleading. Reported wealth is net of consumption, thus the more people react to a given shock by adapting their consumption the smaller is the change in reported wealth. The first stage in such instrumental variable (IV) setup would be biased towards zero, artificially inflating the IV estimate.

However, it is unlikely that people entirely compensate the overall wealth shock between two interviews. Therefore I can test to which extent predicted wealth shocks actually predict changes in reported wealth (this is the 'attenuated first stage'). Figure 4 has already suggested that wave aggregates of reported wealth correlate with the stock market for stock owning retirees but not for those without stocks. I repeat this exercise in Table 4 using individual data instead of wave aggregates. In column 1 and 2 percentage changes in life-time wealth are regressed on the interaction of stock market changes with a stock ownership dummy while controlling for main effects. In column 3 and 4 stock market changes are interacted with the exact fraction of wealth held in stocks (representing the measure of predicted percentage wealth shocks), again controlling for the main effects. As the measurement of stocks in IRAs is noisy in the first waves, I repeat both regressions excluding these waves.

The coefficients on the interaction terms are strongly significant in all four columns. The stock market main effects which represent the effect on the wealth of retirees without stocks is close to zero and -if anything- slightly negative. These results indicate that a 10% change in the stock market leads to an average change of 1%-2% in reported life-time wealth for those with stocks in the previous period. Further, a 10% predicted wealth shock is associated with a change of about 7.7%-8% in reported life-time wealth. This shows that reported wealth is strongly correlated with predicted wealth shocks though

with an elasticity smaller than one which could be interpreted as evidence of consumption smoothing. For those without stocks there is not much of an effect, giving further support to the conclusion from Figure 4, namely that stock market effects through other wealth components are not a big issue.

Finally, choosing retirees as study population has advantages beyond ruling out labour market interactions. Retirees have a lot of wealth and wealth heterogeneity, so there is a lot of variation to exploit. Further it is relatively easy to compute retiree life-time wealth since pension and annuity income, social security and veteran benefits typically stay constant in real terms until death. And the HRS as ideal data set in terms of variables mostly consists of retirees.

4 Empirical Specification

The identification strategy outline above leads to a straight-forward empirical specification.

$$\Delta H_{i,t} = \alpha + \beta \frac{\widehat{\Delta W}_{h(i),t}}{W_{h(i),t-1}} + \gamma \frac{s_{h(i),t-1}}{W_{h(i),t-1}} + \vartheta_t + \delta X_{i,t} + \epsilon_{i,t} \tag{1}$$

with
$$\frac{\widehat{\Delta W}_{h(i),t}}{W_{h(i),t-1}} = \frac{s_{h(i),t-1}}{W_{h(i),t-1}} \frac{SP_{m(i,t)} - SP_{m(i,t-1)}}{SP_{m(i,t-1)}}$$

with indices:

i: individual

t: HRS wave (biannual)

m(i,t): month of the interview of individual (i) in wave (t)

h(i): household of (i)

and variables:

H: different health measures (normalised as explained in the Data section)

SP: Standard & Poor's 500 stock market index

s: stock holdings

W: life-time wealth = $A_t + (SS_t + PAI_t + VetBen_t) \sum_{\tau=1}^{T-t} \frac{S_{t,\tau,g,b}}{(1+\tau)^{t+\tau}}$

 ϑ : time dummies

X: age, age^2 , years of education, dummies for gender (2-1), race (3-1), region of residence (5-1), degree (5-1), past wave's marital status (8-1), and number of months between interviews

Changes in different health measures are directly regressed on the interaction of stock market changes with the fraction of life-time wealth held in stocks. The fraction of wealth in stocks is included separately (person main effect) to guarantee conditional independence of predicted wealth shocks. Year dummies (a flexible specification of stock market main effects) capture common macro shocks that directly affect health. Health measures are regressed in first differences and further controls are included to reduce the regression residual and thereby increase the precision of the estimates. However, these further controls (except for number of months)¹⁸ should not change point estimates if predicted shocks are conditionally independent. Notice that taking the first difference does not reduce the number of observations since past wave's stock holdings and stock market changes already require one lag. Further one must not add controls which could be affected by wealth shocks ('bad controls' in the language of Angrist and Pischke (2009)).

¹⁸, Number of months' are included to control for potential interview delays due to health problems.

the lagged instead of the current marital status is included.

For better comparability the different health measures are normalised (see Data section) and re-coded such that changes are interpreted in terms of standard deviations and positive changes always refer to a health improvement. OLS regressions on these transformed values ('probit-adapted OLS') allow to compare effect sizes across the different health measures, representing effects as moving within the sample distribution rather than across differently scaled categories. Standard errors are clustered by individuals as first differences are likely to be serially correlated. Clustering at the level of households, interview dates or stock market changes leads to lower standard errors.

5 Findings

Table 5 presents the baseline results. Different health measures (rows) are regressed on predicted wealth shocks and a cumulative number of controls (columns). The final column shows results with all controls excluding the first waves for which the fraction of stocks in IRAs is not known. All estimates in this and the following tables display the coefficient on wealth shocks in the different specifications. Positive coefficients refer to health improvements in terms of standard deviations of the different health measures.

The bivariate regressions in the first column indicate a strongly significant relationship of predicted wealth shocks with the health problem index, the self-reported change in health and survival, while there is a weakly significant correlation with self-reported health and no significant relationship with mental health. Including initial stock holdings in the second column does not change this pattern dramatically, indicating that the coefficients in the first column are not driven by average differences of people with different fractions of their wealth held in stocks.¹⁹ When time dummies are included in the third column, all coefficients except for mental health decrease by one half, suggesting that stock market changes come along with macroeconomic shocks which also affect the health of those without stocks. The coefficient in the mental health regression, on the other hand, increases and becomes significant. This might indicate that the mental health of those without stocks is reversely affected by stock market changes while there is not much of an effect for stock owners. However, repeating the first two regressions in the sample restricted to vear>1999 results in significant positive effects for stock owners. Hence the insignificant results when including the first waves should not be overinterpreted. Adding further additional controls in column 4 seems to increase the precision of the estimates in some cases but does not change the overall pattern. This is what we should expect if wealth shocks conditional on stock holdings are truly exogenous. Excluding waves 1992-1996 in column 5 (we are left with waves 2000-2009 as the 1998 wave is used as a lag) changes coefficients

¹⁹This is not surprising as for the predicted wealth shocks these stock holdings are interacted with positive as well as negative stock market changes, so that any bias would be reverted if interacted with negative changes. Also the dependent variable is regressed in first differences canceling out any individual fixed effect driving stock holdings.

somehow but again the overall pattern remains the same. The fact that in column 5 more precise information on stocks in IRAs is used (for the first waves an average of 52% stock share for IRA is assumed) should increase the precision of the estimates (as explained in the Data section). But at the same time the sample size is decreased, driving up standard errors.

To sum up, the baseline regressions in Table 5 suggest that the predicted wealth shocks have a significant positive effect of similar size on the health problems index, self-reported health, mental health and mortality. Coefficients in column 5 indicate that a +10% shock in life-time wealth is associated with an increase of about 1.5% to 3% of a standard deviation in the different health measures. These effects are not driven by person nor by time effects and therefore can be interpreted as the causal effect of wealth shocks on health. The inclusion of various controls does not affect the overall pattern nor the exclusion of the first waves.

The baseline regression setup relates health changes to wealth shocks over a one to three year period. If the estimated effects represent causal effects of wealth on health, they must be driven by diseases that are responsive to environmental factors and that do not take a lot of time to develop.

To get an idea of which health problems are driving the estimated effect. Table 6 shows results for the decomposed health problem index, using the individual health problems of the index as the dependent variables in separate regressions. Estimates refer to the coefficients of wealth shocks in the full specification (as Table 5, column 4 and 5). As in the previous regressions, each dependent variable is normalised such that positive coefficients indicate a health improvement (i.e. a lower chance to get a new health problem) in terms of standard deviations. The regressions reveal a strong and positive effect of wealth shocks on the incidence of high blood pressure (ie. a negative effect on the probability to get high blood pressure) and effects of about half the size for heart diseases, psychiatric problems and - surprisingly - cancer; in specifications with less 'other controls' the coefficient for strokes is significant, too. The incidence of arthritis, diabetes and lung disease, on the other hand, is not affected. These heterogeneous effects across different diseases seem reasonable. High blood pressure and psychiatric problems are plausible drivers for a short to medium term effect of wealth shocks on the health of wealthy retirees that are covered by health insurance (medicare). High blood pressure arises from both psychological stress as well as unhealthy nutrition and behavior. Moreover, high blood pressure is a cause for heart problems (and strokes), but not for diabetes, cancer, lung diseases or arthritis. Therefore a positive effect on heart problems and strokes is what one should expect given the strong effect on high blood pressure. The effect of wealth shocks on cancer is in line with effect of stress on cancer incidence that is supposed by the epidemiological literature.²⁰ When the first waves are excluded in the second column standard errors go up and

 $^{^{20}}$ Note that those with positive wealth shocks might smply tend to do better cancer checkups which increases the likelihood of cancer diagnosis even if the risk of cancer might be lower than for those with

all coefficients except the one on high blood pressure become insignificant. The central message from this table is that wealth shocks have a strong effect on high blood pressure, a small effect on diseases which are caused by high blood pressure as well as on cancer, but no effect on effect on arthritis, diabetes and lung disease. Further it seems that large sample sizes are needed to detect these smaller effects.

Table 7 shows results for the individual items of the mental health index, which indicate the absence of the respective negative emotion or the presence of the respective positive emotion "over much of the time during the past week". Hence a positive change indicates the appearance of a positive emotion or the disappearance of a negative emotion compared to the previous interview, and vice versa. Results indicate weakly significant effects on feeling depressed and lonely, while the coefficients on the other items are smaller but still in all cases the right direction. No single item seems to be the single driver of the effect on the overall index, as one should expect. The mental health index does not - unlike the health problems index - represent a list of different diseases but a collection of symptoms associated with clinical depression. Any single symptom is not necessarily a sign of depression but what makes it a mental health problem is having many of the symptoms at the same time. Therefore it is not surprising that the significant estimate for the mental health index is not driven by single items but rather represents their joint significance.

How big are the estimated effects? To answer this question it is insightful to compare them with a simple benchmark regression of the respective health measure on wealth and some controls. The resulting coefficient will be driven not only by the causal effect of wealth on health but also by reverse causality and any simultaneous effect from omitted variables. As one would expect reverse causality and omitted variables to have a positive influence, the coefficient from such benchmark regression should be larger than the estimated causal effect.

The equivalent to the predicted wealth shocks are percentage changes in reported lifetime wealth (used as dependent variable in Table 4). Table 14 shows in column 1 the estimated causal effects from the baseline regressions and in column 2 the coefficients from a regression of health changes on relative changes in reported wealth and age, gender and race as controls.²¹ Surprisingly the coefficients in column 2 are very small and not even significantly different from zero for the health problems index, the mental health index and survival. An explanation for smaller estimates could be attenuation due to measurement error. Self-reported income and especially wealth are notorious for measurement errors. And as retirement income is quite stable and wealth is regressed in first differences, much of the variation might represent mere measurement error biasing the estimates towards

negative shocks. However, one would expect to see such checkup effect also for other diseases like high blood pressure.

²¹Notice that in the baseline regressions additional controls (besides the main effects) are included only to improve the precision of the estimates, as predicted wealth shocks should be conditionally independent of these controls. This is probably not true for wealth in the benchmark regression, Therefore only a few, clearly exogenous controls are included. However, if more controls are added, benchmark estimates become even smaller.

zero (Angrist and Pischke 2009, chapter 5.1.).

Potential measurement errors are less of a problems in a level-regression. The levelequivalent to the regression in column 2 is the regression of health levels on the logarithm of life-time wealth.²² The resulting coefficients in column 3 are much larger than in the second column and significantly positive in all cases. Notice that this increase might not only be due to less measurement error but probably partly caused by fixed effects that were canceled out in first differences. However, estimates are still only slightly larger than the estimated causal effects and even smaller in the survival regression. This seems to be at odds with a positive bias due to reverse causality and omitted variables.

One explanation could be that predicted wealth shocks underestimate actual wealth shocks due to non-stock wealth that correlates with the stock market. But in this case reported wealth of those without stocks should correlate positively with the S&P500 which is not the case as shown above (see Tables 3 and 4). Another reason could be that life-time wealth is overestimated, which would make predicted wealth shocks (which are rescaled by life-time wealth) appear smaller than they actually are. However, artificially underestimating the life-time wealth formula, e.g. even by assuming that people live only one more year does not alter a lot the size of the estimated causal effect relative to the respective benchmark regression.

A more plausible explanation is that predicted wealth shocks correctly represent actual wealth shocks but the resulting causal effect on health might not be representative for the average causal effect of wealth on health in the sample. Much of the causal effect of wealth on health in the benchmark regression should come from the long-term wealth elasticity of health. Someone owning \$500k can afford better health inputs than somebody owning \$300k which over time accumulates to better health outcomes. This might however be a very different effect from losing \$200k in a stock market crash, which involves high blood pressure, stress and depression rather than just a slight change in health inputs.

Analysing individual health problems in Table 9 provides evidence in favor of such wealth shock interpretation. While the estimated causal effect of wealth shocks differs across health problems (between zero and 0.17), the benchmark relationship is quite homogeneous, ranging from 0.05 to 0.1 (except for cancer, see below). This means that for high blood pressure the benchmark relationship amounts to only one third of the estimated causal effect while it is of similar size for heart disease and psychiatric problems. For other health problems like arthritis and diabetes the estimated causal effect is zero but the correlation is significantly positive (and hence larger). Cancer stands out with a negative wealth relationship, indicating that richer people are less protected from cancer diagnosis (controlling for age, gender and race), while the estimated causal effect is positive. These

²²Percentage changes are equivalent to the logarithm only for small changes and there are wealth shocks up to +/-50% included. However, regressing predicted wealth shocks as logarithmic instead of percentage changes (i.e. $ln(\widehat{W}_t/W_{t-1})$ instead of $\widehat{\Delta W_t}/W_{t-1}$) results in very similar estimates.

results suggest that the estimated effects of wealth shocks are indeed of a different quality and not just an inflated proxy for the average causal effect of wealth in the sample.

Tables 10 - 13 present various robustness checks, again using the baseline specification for better comparability. Table 10 compares the baseline results with IV regressions in which predicted wealth shocks are instrumented by the 1992-2006 average stock holdings (column 3), the 1998-2006 average (column 4) and initial 1998 stock holdings (column 5). These instruments are uncorrelated with where we are in the stock market cycle and therefore rule out a potential endogeneity due to stock market predictability (see Identification section). For the health problems index and for mental health coefficients become slightly larger in the IV setup while they slightly decrease in the other regressions. Standard errors increase throughout, which is what one should expect.²³ Given that coefficients only change slightly and mostly remain significant despite this increase in noise suggests that the baseline findings are not driven by stock market predictability.

Specification checks in Table 12 indicate that results are not driven by the comparison of wealthy stock owning retirees with those who are poor or who do not own stocks. When excluding the bottom life-time quartile or those without stocks, coefficients only change slightly and remain mostly significant despite increasing standard errors due to decreased sample size. In the last two columns I repeat the baseline regressions using a simplified life-time formula representing the sum of current wealth and only one year of pension income (instead the discounted sum of all future expected income streams). Estimated coefficients decrease since predicted wealth shocks are rescaled by a smaller life-time wealth, but significance levels and the overall pattern do not change.

Heterogeneity checks in Table 13 show that effects are quite homogeneous across age, gender and negative vs. positive shocks. Only for mortality effects are significantly different between subsamples. Mortality for retirees above age 75 is stronger affected than for those below 75. This seems plausible given that the elderly are closer to the margin of death while this is not the case for mental health or self-reported health. Further negative wealth shocks seem to have a more devastating impacts on mortality than the positive effects of positive shocks of equal size. This could be in line with concepts like loss aversion, though it remains unclear why the same pattern is not observed for the other health measures.

6 Conclusion

This paper investigates the effect of wealth shocks on health by regressing health changes over a two-year period on stock market induced wealth shocks over the same period. I find

 $^{^{23}}$ In both IV-setups information is lost by using proxies instead of actual values for stock holdings. And in column (5) I am additionally losing many observations, all those who do not report stock holdings in 1998, in particular the cohorts that were added in 2004.

significant positive effects of similar size on physical health, self-reported health, mental health and mortality. A 10% change in life-time wealth is associated with a change of about 1.5-3% of a standard deviation in these health measures.

Effects are heterogeneous across different physical health problems, with most pronounced effects for the incidence of high blood pressure, smaller effects for heart problems, strokes and cancer, and no effects for arthritis, diabetes and lung diseases. This pattern suggests that psychological factors play an important role in the transmission of effects from wealth shock to retire health. Estimated effect sizes are of equal size or larger than the overall wealth-health relationship. This suggests that effects are larger than the average causal effect of wealth on health in the sample. Effects are quite homogenous across age, gender and types of wealth shocks. Only for the mortality regressions there is a significantly stronger effect for the elderly and if wealth shocks are negative.

There are three main conclusions. First, wealth shocks have a causal effect on physical health, mental health and mortality of wealthy retirees in the US. Second, effects too large to represent the average causal effect of wealth on health in the sample. Third, psychological factors seem be a major channel through which these effects.

7 References

Adams, Peter, Michael D. Hurd, Daniel McFadden, Angela Merrill and Tiago Ribeiro (2003) 'Healthy, Wealthy, and Wise? Tests for Direct Causal Paths between Health and Socioeconomic Status.' *Journal of Econometrics*. January, 112:1, pp. 3-56.

Angrist, Joshua D. and Jörn-Steffen Pischke (2009). *Mostly Harmless Econometrics: An Empiricist's Companion*, Princeton, NJ: Princeton University Press.

Apouey, Benedicte and Andrew Clark (2009). Winning Big but Feeling No Better? The Effect of Lottery Prizes on Physical and Mental Health' mimeographed, PSE (December).

Case, Anne, Darren Lubotsky, and Christina Paxson. (2002). Economic status and health in childhood: the origins of the gradient. *American Economic Review* 92/5: 1308-1334.

Case, Anne (2004). 'Does Money Protect Health Status? Evidence from South African Pensions' in David Wise (Ed.), *Perspectives on the Economics of Aging*. Chicago: University of Chicago Press, 2004.

Chida, Yoichi, Mark Hamer, Jane Wardle, and Andrew Steptoe (2008). 'Do stress-related psychological factors contribute to cancer incidence and survival: a systematic review and meta-analysis.' *Nature Clinical Practice Oncology*; 5:46675.

Chida, Yoichi, and Andrew Steptoe (2008). Positive psychological wellbeing and mortality: a quantitative review of prospective observational studies. *Psychosomatic Medicine*, 70:741-756.

Cohen, Sheldon, Cuneyt M. Alper, William J. Doyle, John J. Treanor, and Ronald B. Turner (2006). 'Positive emotional style predicts resistance to illness after experimental exposure to rhinovirus or influenza a virus.' *Psychosomatic Medicine*, 68:809-815.

Coile, Courtney C. and Phillip B. Levine (2006). 'Bulls, Bears, and Retirement Behavior.' Industrial and Labor Relations Review 59(3):408-429.

Coile, Courtney, Peter Diamond, Jonathan Gruber, and Alain Jousten (2002). 'Delays in Claiming Social Security Benefits, *Journal of Public Economics*, 84(3), pp. 357-85.

Cutler, David M. (2005). Comment on Smith, James P. (2005). 'Consequences and Predictors of New Health Events' in David Wise (Ed.) Advances in the Economics of Aging. University of Chicago Press, 2005.

Cutler, David M., Deaton, Angus, Lleras-Muney, Adriana (2006). 'The Determinants of Mortality.' *Journal of Economic Perspectives*, 20(3).

Cutler, David M., Lleras-Muney, Adriana, and Vogl, Tom (2010). 'Socioeconomic Status and Health: Dimensions and Mechanisms', in *Oxford Handbook of Health Economics*, forthcoming.

Deaton, Angus (2003). Health, inequality, and economic development, *Journal of Economic Literature* 41/1: 113-158.

Deaton, Angus and Christina Paxson (2001). 'Mortality, Education, Income, and Inequal-

ity among American Cohorts', in David Wise (Ed.), *Themes in the Economics of Aging*. Chicago U. Press for NBER (NBER work. paper 7140)

Deaton, Angus and Christina Paxson (2004). Mortality, income, and income inequality over time in Britain and the United States, in David Wise (Ed.), *Perspectives on the Economics of Aging.* Chicago: University of Chicago Press, 247-279.

Duflo, Esther (2000). 'Child Health and Household Resources in South Africa: Evidence from the Old Age Pension Program.' American Economic Review 90(2):39398.

Engelhardt, Gary V. and Jonathan Gruber (2005), .Social Security and Elderly Well-Being in the United States, mimeographed, MIT (May).

Easterlin, Richard A. (1974). 'Does Economic Growth Improve the Human Lot? Some Empirical Evidence' in: David, Paul A. and Reder, Melvin W., Editors, *Nations and Households in Economic Growth: Essays in Honour of Moses Abramovitz*, Academic Press, Inc., New York.

Frijters, Paul, John P. Haisken-DeNew, and Michael A. Shields (2004). 'Money Does Matter! Evidence from Increasing Real Income and Life Satisfaction in East Germany Following Reunification' *American Economic Review*, 94, No. 3 (June), 730.740.

Gardner, Jonathan and Andrew J. Oswald (2007). 'Money and mental wellbeing: A longitudinal study of medium-sized lottery wins' *Journal of Health Economics*, 26, No. 1 (January), 49-60.

Goda, Gopi Shah, John B. Shoven, and Sita Nataraj Slavov (2010). 'Does Stock Market Performance Influence Retirement Expectations?', *NBER Working Paper*

Hurd, Micheal D., Monika Reti and Susann Rohwedder (2009). 'The Effect of Large Capital Gains or Losses on Retirement.' in , David Wise (Ed.), *Developments in the Economics* of Aging. Chicago: University of Chicago Press.

Jensen, Robert and Kaspar Richter (2003). 'The health implications of social security failure: Evidence from the Russian pension crisis.' *Journal of Public Economics*, 88.

Lindahl Mikael (2005). 'Estimating the effect of income on health and mortality using lottery prizes as an exogenous source of variation in income' *Journal of Human Resources*; 40; 144-168.

Malkiel, Burton G., (2003). 'The efficient market hypothesis and its critics' *Journal of Economic Perspectives* 17, 5982.

Mealli, Fabrizia and Donald Rubin (2003). 'Commentary: Assumptions Allowing the Estimation of Direct Causal Effects' *Journal of Econometrics*, 112, pp. 79-87.

Michaud, Pierre-Carl, and Arthur Van Soest (2008). Health and Wealth of Elderly Couples: Causality Tests using Dynamic Panel Data Models. *Journal of Health Economics*, 27(5): 131225.

Pischke, Jörn-Steffen (2010) 'Money and Happiness: Evidence from the Industry Wage Structure'

Ruhm, Christopher J. (2000). Are recessions good for your health? *Quarterly Journal of Economics* 115/2: 617-650.

Smith, James P. (1999). Healthy bodies and thick wallets: The dual relation between health and economic status. *Journal of Economic Perspectives* 13/2: 145-166.

Smith, James P. (2005). 'Consequences and Predictors of New Health Events' in David Wise (Ed.) Advances in the Economics of Aging. University of Chicago Press, 2005.

Snyder, Stephen S., and William N. Evans. (2006). The impact of income on mortality: evidence from the Social Security notch. *Review of Economics and Statistics* 88/3: 482-495.

Strike, Philip C. and Andrew Steptoe (2004). 'Psychosocial Factors in the Development of Coronary Artery Disease' *Progress in Cardiovascular Diseases*, Vol. 46, No. 4, 337-347

Sullivan, Daniel and Till von Wachter (2009). 'Job Displacement and Mortality: An Analysis using Administrative Data' *Quarterly Journal of Economics*, 124(3), 1265-1306.

van Praag, Bernard M.S. and Ada Ferrer-i-Carbonell (2008) *Happiness Quantified: A Satisfaction Calculus Approach*, revised edition, Oxford: Oxford University Press

8 Tables and Figures

HRS wave	~	2	З	4	5	9	7	8	6
year	1992-1993	1994-1995	1996-1997	1998-1999	2000-2001	2002-2003	2004-2005	2006-2007	2008-2009
# observations	12,605	11,389	11,150	22,419	20,770	19,521	21,349	19,676	18,434
age	55	57	59	67	68	69	67	69	70
% retiree households	22%	30%	38%	63%	66%	20%	64%	68%	%02
information '% of IRA in stocks'	none	none	none	3 categories	3 categories	3 categories	3 categories	exact %	exact %
imputed % of IRA in stocks	52%	52%	52%	0, 50, 100%	0, 50, 100%	0, 50, 100%	0, 50, 100%	exact %	exact %
regression sample									
# observations		941	1,899	2,611	8,334	7,910	8,532	8,464	7,942
current wealth	198,660	244,801	278,882	272,641	320,652	343,968	395,071	487,048	494,830
life-time wealth	416,623	541,103	591,140	490,127	558,341	595,344	665,405	774,073	863,006
% with stocks (excl IRAs)	29%	34%	33%	31%	33%	31%	30%	28%	27%
% with stocks (incl IRAs)	38%	46%	47%	32%	34%	33%	32%	31%	30%
those with stocks (incl. IRAs)									
# observations	·	401	975	1,369	3,089	3,125	3,181	3,216	2,872
current wealth	375,552	404,323	455,005	520,935	580,899	625,063	769,296	927,367	929,626
life-time wealth	650,643	780,726	842,988	795,207	873,734	936,483	1,098,133	1,277,645	1,343,069
stock holdings (incl IRAs)	89,232	98,677	129,586	195,696	220,003	199,866	301,739	356,956	319,659
% life-time wealth in stocks	11%	12%	12%	19%	20%	19%	20%	20%	21%
Notes: New cohorts added in 1998 and	d 2004; retiree	households r	efer to singles	s and couples	where both r	espondent an	d spouse -if e	xistent- were	neither

Table 1: Summary Statistics Wealth

working nor unemployed; the regression sample includes all households that in the previous wave were retired and reported their wealth, retiree income (for the construction of life-time wealth) and stock holdings - hence there are no observations in wave 1 and the wealth information in wave 1 refers to the num-ber of observations in wave 2; life-time wealth is the sum of current wealth and future discounted retiree income (see Data section).



Figure 1: S&P500 with indicated HRS observation periods

Figure 2: Distribution of Changes in S&P500



The distribution of percentage changes in the S&P500 between two interviews of the same household is displayed. Values around -0.4, for example, refer to interviews that were conducted in wave 6 or 9 at a time when the S&P500 was 40% below its value at the households's previous interview.



Figure 3: Predicted Wealth Changes and the S&P500 over Time

Predicted wealth shocks ('past interview's wealth fraction held in stocks'*'percentage change in the S&P500') over time are plotted with the S&P500. Each circle represents one household and is placed in the figure at the exact month of the household's interview.

HRS wave	~	7	ო	4	5	9	7	8	6	
year	92 93	94 95	96 97	98 99	00 01	02 03	04 05	06 07	08 09	
# observations	12,605	11,389	11,150	22,419	20,770	19,521	21,349	19,676	18,434	
# deaths	י י	45 12	4 437 488	542 637	574 615	566 555	592 581	569 566	343 -	
age at death	, ,	61 6	3 78 79	79 79	78 79	80 81	80 80	81 80	81 -	
regression sample										
# observations included	'	873	1,802	2,468	7,954	7,625	8,210	8,171		
# deaths included	1 1	8 17	26 32	42 70	154 414	325 395	343 437	335 425	242 -	
Notes: Deaths are regressed as a 'Surv interview and 1 otherwise. This survival regression sample one lag and one futu and wave 9. The final row shows how <i>m</i> included death, the deceased responde interview. For example, deaths in 1994. but in 1996 a proxy respondent reported	ival' dumry dummy is re observ nany deat nat was re are incluc	ny that in s regresse ation is re ns in each irred and i the i individual	dicates surv ed on wealth aquired. For t year are in eported her ndividual we died in 199.	ival to t+1, shocks be that reasor cluded in th wealth an is retired a t (obviousl	i.e. it is 0 it tween t-1 a there are a sample d health in nd reporte y after the	the indivi and t, so the no regres (accounte t-1 and t, d wealth a interview).	dual desea nat for eacl sion obser d by surviv and decea nd health ii	ises before h observati vations in v vations in v vations in v vations in v ations in v 1992 anc	the next on in the vave 1 each the t+1 1994,	

Table 2: Summary Statistics Deaths in the HRS



Figure 4: S&P500 and Reported Wealth Changes over Time

Coefficients from Table 3, columns 1 and 2 are plotted; $\Delta A(t)$ = changes in reported current wealth.

	Dep. Var	$\therefore \Delta A_t$
	without stocks in t-1	with stocks in t-1
wave2	4,837	$44,\!625$
	(5,170)	(29,007)
wave3	8,214	$30,839^{**}$
	(7, 364)	(15, 246)
wave4	$5,\!660$	$13,\!687$
	(4,011)	(17, 191)
wave5	11,406***	51,491***
	(2,566)	(15, 246)
wave6	$20,365^{**}$	-40,480**
	(9,155)	(17,881)
wave7	14,041	67,154*
	(9,656)	(35,675)
wave8	$26,665^{***}$	117,832**
	(4,056)	(59, 493)
wave9	4,258	-41,819
	(8,111)	(51,207)
n	34,6	26

 Table 3: Reported Wealth Changes over Time

 A_t =reported current wealth; coefficients on wave dummies interacted with an indicator of stock ownership in t-1 are displayed; no constant included; these coefficients are plotted in Figure 4.

Table 4: Regressing Wealth Changes on Stock Market Changes (column 1, 2) and on Predicted Wealth Changes (3, 4).

	(1)	(2)	(3)	(4)
Dep. Var.: $\Delta W_t / W_{t-1}$	full sample	year > 1999	full sample	year>1999
$\Delta SP_t/SP_{t-1}^*D$ stocks _{t-1}	0.133^{***} (0.040)	0.222^{***} (0.054)		
Dstocks _{t-1}	-0.040***	-0.036***		
$\Delta SP_t / SP_{t-1}^* [s_{t-1} / W_{t-1}]$	(0.013)	(0.013)	0.777***	0.800***
s_{t-1}/W_{t-1}			(0.122) - 0.222^{***}	(0.148) - 0.216^{***}
$\Delta SP_t/SP_{t-1}$	-0.044	-0.038	(0.041) -0.047*	(0.042) -0.018
constant	(0.031) 0.174^{***}	(0.040) 0.169^{***}	(0.025) 0.176^{***}	(0.034) 0.171^{***}
	(0.009)	(0.009)	(0.008)	(0.008)
n	$34,\!619$	$30,\!926$	$34,\!619$	30,926

 W_t = life-time wealth (see 'Data' section); SP_t = S&P500 stock market index; Dstocks_{t-1}=(stockholdings_{t-1} > 0); s_{t-1}=stockholdings_{t-1}; Sample: households retired in t-1; 3 observations with $\Delta \% W_t$ > 5000% excluded; (standard errors) clustered by household.

Dep. Var.	(1)	(2)	(3)	(4)	(5)
Δ Index of Health Problems n=36,502	$\begin{array}{c} 0.256^{***} \\ (0.045) \end{array}$	$\begin{array}{c} 0.253^{***} \\ (0.047) \end{array}$	$\begin{array}{c} 0.154^{***} \\ (0.053) \end{array}$	$\begin{array}{c} 0.149^{***} \\ (0.053) \end{array}$	$\begin{array}{c} 0.166^{***} \\ (0.060) \\ 33,870 \end{array}$
Δ Self-reported Health n=46,220	0.162^{*} (0.083)	0.263^{***} (0.088)	0.106 (0.100)	0.121 (0.100)	$\begin{array}{c} 0.211^{*} \\ (0.122) \\ 40,827 \end{array}$
Self-reported Change in Health n=46,244	$\begin{array}{c} 0.775^{***} \\ (0.090) \end{array}$	0.665^{***} (0.091)	$\begin{array}{c} 0.358^{***} \\ (0.102) \end{array}$	$\begin{array}{c} 0.338^{***} \\ (0.101) \end{array}$	$\begin{array}{c} 0.215^{*} \\ (0.114) \\ 40,849 \end{array}$
Δ Mental Health Index n=38,932	-0.135 (0.101)	-0.088 (0.106)	0.218^{*} (0.120)	0.233^{*} (0.120)	$\begin{array}{c} 0.312^{**} \\ (0.144) \\ 36{,}550 \end{array}$
Survival n=36,804	$\begin{array}{c} 0.391^{***} \\ (0.062) \end{array}$	0.280^{***} (0.067)	0.141* (0.079)	$\begin{array}{c} 0.154^{**} \\ (0.073) \end{array}$	$\begin{array}{c} 0.204^{**} \\ (0.090) \\ 31,710 \end{array}$
controls					
past wave's stock fraction time dummies other controls restricted to year>1999		V	√ √	\checkmark \checkmark	\checkmark

Table 5: Baseline Regressions

The coefficient on wealth shocks $(\widehat{\Delta W}_t/W_{t-1})$, equation 1) is displayed; a positive coefficient refers to a health improvement in the respective dependent variable; all dependent variable (including 'Survival') are standardized as explained in the data section; $\Delta X = X_t - X_{t-1}$; 'Surival' is 0 if respondent deceases before (t+1) and 1 otherwise; the estimation method used is OLS ('probit-adapted' OLS); standard errors, in parethesis, are clustered by individual; 'other controls' are: age, age², years of education, dummies for gender, race, initial marital status, degree and # months since previous interview.

Dep. Var.	(1)	(2)	
Δ Health Problem Index	0.149^{***} (0.053)	0.166^{***} (0.060)	
	decomposed		
Δ high blood pressure	0.168***	0.169^{***}	
0	(0.050)	(0.058)	
Δ heart disease	0.098**	0.082	
	(0.050)	(0.060)	
Δ stroke	0.073*	0.071	
	(0.043)	(0.053)	
Δ arthritis	0.023	0.037	
	(0.057)	(0.064)	
$\Delta cancer$	0.081^{*}	0.071	
	(0.044)	(0.053)	
Δ diabetes	0.010	-0.037	
	(0.031)	(0.036)	
Δ lung disease	0.039	0.036	
	(0.032)	(0.035)	
Δ psychiatric problems	0.079^{**}	0.058	
	(0.036)	(0.043)	

Table 6: Individual health problems

The coefficient on wealth shocks $(\widehat{\Delta W}_t/W_{t-1})$, equation 1) is displayed; a positive coefficient refers to a health improvement in the respective dependent variable; all dependent variables are standardized as explained in the data section; $\Delta X = X_t - X_{t-1}$; the estimation method used is OLS ('probit-adapted' OLS); standard errors, in parethesis, are clustered by individual; 'other controls' are: age, age², years of education, dummies for gender, race, initial marital status, degree and # months since previous interview.

Dep. Var.	(1)	(2)	
AMental Health Index	0.233*	0 319**	
	(0.120)	(0.144)	
	decomposed		
Δ felt depressed	0.193^{*}	0.194	
	(0.102)	(0.124)	
Δ everything is an effort	0.162	0.095	
	(0.107)	(0.128)	
Δ sleep is restless	0.082	0.131	
	(0.123)	(0.148)	
Δ felt alone	0.193^{*}	0.251**	
	(0.105)	(0.126)	
Δ felt sad	0.155	0.204	
	(0.113)	(0.133)	
Δ could not get going	0.048	0.101	
	(0.118)	(0.141)	
Δ felt happy	0.041	0.055	
	(0.101)	(0.120)	
Δ enjoyed life	0.053	0.132	
	(0.088)	(0.104)	
controls			
past wave's stock fraction	n 🗸	\checkmark	
time dummies	\checkmark	\checkmark	
other controls	\checkmark	\checkmark	
restricted to year>1999		\checkmark	

Table 7: Individual items of the mental health index

The coefficient on wealth shocks $(\widehat{\Delta W}_t/W_{t-1})$, equation 1) is displayed; a positive coefficient refers to a health improvement in the respective dependent variable; all dependent variables are standardized as explained in the data section; $\Delta X = X_t - X_{t-1}$; the estimation method used is OLS ('probit-adapted' OLS); standard errors, in parethesis, are clustered by individual; 'other controls' are: age, age², years of education, dummies for gender, race, initial marital status, degree and # months since previous interview.

	(1)	(2)	(3)
	ΔH_t on	ΔH_t on	H_t on
	$\widehat{\Delta W}_t / W_{t-1}$	$\Delta W_t / W_{t-1}$	lnW_t
Dep. Var.: H_t measure	(Table 5, (4))		
Health Problems Index	0.149^{***}	0.001	0.177^{***}
	(0.053)	(0.002)	(0.009)
SRH	0.121	0.018^{***}	0.286^{***}
	(0.100)	(0.005)	(0.007)
$SR\Delta H$	0.338^{***}	0.015^{***}	-
	(0.101)	(0.004)	-
Mental Health	0.233^{*}	0.005	0.230^{***}
	(0.120)	(0.005)	(0.007)
Survival	0.154^{**}	-0.002	0.054^{***}
	(0.073)	(0.004)	(0.003)
controls			
past wave's stock fraction	\checkmark		
time dummies	\checkmark		
other controls	\checkmark		
age, age^2 , gender, race		\checkmark	\checkmark

Table 8: Comparison with unconditional health-wealth relationship in the sample

Coefficients on $\widehat{\Delta W}_t/W_{t-1}$ (column 1), $\Delta W_t/W_{t-1}$ (2), and lnW_t (3) are displayed; all waves included; (2) and (3) are estimated for the same sample as (1); there is no estimate for $SR\Delta H$ in (3) as $SR\Delta H$ is only defined in first differences.

	(1)	(2)
	ΔH_t on	H_t on
	$\widehat{\Delta W}_t / W_{t-1}$	lnW_t
Dep. Var.: H_t measure	(Table 6, (1))	
Health Problem Index	0.149^{***}	0.177^{***}
	(0.053)	(0.009)
decom	nposed	
high blood pressure	0.168^{***}	0.057^{***}
	(0.050)	(0.007)
heart disease	0.098^{**}	0.060^{***}
	(0.050)	(0.007)
stroke	0.073^{*}	0.058^{***}
	(0.043)	(0.005)
arthritis	0.023	0.051^{***}
	(0.057)	(0.006)
cancer	0.081^{*}	-0.040***
	(0.044)	(0.006)
diabetes	-0.010	0.081^{***}
	(0.031)	(0.006)
lung disease	0.039	0.077^{***}
	(0.032)	(0.005)
psychiatric problems	0.079^{**}	0.103^{***}
	(0.036)	(0.006)
controls		
past wave's stock fraction	\checkmark	
time dummies	\checkmark	
other controls	\checkmark	
$age, age^2, gender, race$		✓

Table 9: Unconditional health-wealth relationship for individual health problems

Coefficients on $\widehat{\Delta W}_t/W_{t-1}$ (column 1) and lnW_t (2) are displayed; column 1 repeats the baseline results of Table 6; (2) is estimated for the same sample as (1).

Table 10: Instrumenting predicted wealth shocks $\left(\frac{\Delta S\&P_t}{S\&P_{t-1}} * \frac{s_{t-1}}{W_{t-1}}\right)$ by the interaction of S&P500 changes with the individual's average stock fraction $\left(\frac{\Delta S\&P_t}{S\&P_{t-1}} * \left[\frac{s}{W}\right]^{average}\right)$ or the individual's 1998 stock fraction $\left(\frac{\Delta S\&P_t}{S\&P_{t-1}} * \left[\frac{s}{W}\right]^{1998}\right)$

	(10010 0)	- ' LW I		•• [W]
(1)	(2)	(3)	(4)	(5)
0.153^{***}	0.169^{***}	0.228***	0.243***	0.248^{***}
(0.053)	(0.060)	(0.064)	(0.072)	(0.087)
36,320	33,701	$36,\!270$	$33,\!663$	$23,\!033$
0.121	0.211^{*}	0.019	0.117	0.002
(0.100)	(0.122)	(0.118)	(0.145)	(0.183)
46,220	40,827	$46,\!143$	40,773	$28,\!153$
0.000***	0.015*	0.000**	0.104	0.004
0.338***	0.215*	0.269**	0.104	0.064
(0.101)	(0.114)	(0.119)	(0.129)	(0.157)
46,244	40,849	46,168	40,795	$28,\!170$
0.233*	0.312**	0.266^{*}	0.385**	0.403*
(0.120)	(0.144)	(0.144)	(0.173)	(0.208)
38,932	36,550	38,904	36,522	24,965
0.154^{**}	0.204^{**}	0.117	0.184^{*}	0.078
(0.073)	(0.090)	(0.086)	(0.105)	(0.117)
$36,\!804$	31,710	36,746	$31,\!670$	$23,\!340$
<u> </u>	×			
•	•	\checkmark	\checkmark	
		·	·	\checkmark
\checkmark	\checkmark	\checkmark	\checkmark	√
√				
·	\checkmark	-	\checkmark	\checkmark
	(1) (1) (1) (1) (1) $(0.153)^{***}$ (0.053) (0.121) (0.100) (0.120) (0.120) (0.120) (0.120) (0.120) $(0.154)^{**}$ (0.073) $(0.154)^{**}$ $(0.154)^{**}$ $(0.154)^{**}$ $(0.154)^{**}$	$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(1) (2) (3) (4) 0.153*** 0.169*** 0.228*** 0.243*** (0.053) (0.060) (0.064) (0.072) 36,320 33,701 36,270 33,663 0.121 0.211* 0.019 0.117 (0.100) (0.122) (0.118) (0.145) 46,220 40,827 46,143 40,773 0.338*** 0.215* 0.269** 0.104 (0.101) (0.114) (0.119) (0.129) 46,244 40,849 46,168 40,795 0.233* 0.312** 0.266* 0.385** (0.120) (0.144) (0.144) (0.173) 38,932 36,550 38,904 36,522 0.154** 0.204** 0.117 0.184* (0.073) (0.090) (0.086) (0.105) 36,804 31,710 36,746 31,670 Image: Value V

The coefficient on predicted wealth shocks $(\widehat{\Delta W}_t/W_{t-1} \equiv \frac{\Delta S\&P_t}{S\&P_{t-1}} * \frac{s_{t-1}}{W_{t-1}})$ is displayed; the estimation method used is OLS in columns 1 and 2, and 2SLS in columns 3-5; the IV in (3) is $\frac{\Delta S\&P_t}{S\&P_{t-1}} * [\frac{s}{W}]_{92-06}^{average}$, in (4) is $\frac{\Delta S\&P_t}{S\&P_{t-1}} * [\frac{s}{W}]_{98-06}^{average}$, and in (5) is $\frac{\Delta S\&P_t}{S\&P_{t-1}} * [\frac{s}{W}]^{1998}$. Further comments as in Table 5.

		1		
Dep. Var.	(1) baseline	(2) excl. if $age < 65$	(3) excl. if $s_{t-1} = 0$	(4) excl. poorest quartile
Δ Index of Health Problems	0.149^{***}	0.186^{***}	0.108*	0.146^{***}
	(0.053)	(0.069)	(0.064)	(0.055)
п	36,502	25,879	14,651	28, 225
A Salfranovtad Haalth	0 191	906 0	0.146	0.103
	(0.100)	(0.141)	(0.119)	(0.103)
п	(46, 220)	31,520	(18,072)	34,894
Self-reported Change in Health	0.338^{***}	0.157	0.289^{**}	0.220^{**}
	(0.101)	(0.128)	(0.122)	(0.106)
n	46,244	31,544	18,074	34,904
Δ Mental Health Index	0.233^{*}	0.264	0.215	0.218^{*}
	(0.120)	(0.163)	(0.144)	(0.126)
n	38,932	27,918	15,754	30,246
Survival	0.154^{**}	0.197^{*}	0.167^{*}	0.149^{**}
	(0.073)	(0.109)	(0.088)	(0.076)
n	36,804	24,149	14,696	27,604
controls:				
past wave's stock fraction	>	>	>	>
time dummies	>	>	>	>
other controls	>	>	>	>

Table 11: Alternative sample specifications

Comments as in Table 5.

Dep. Var.	(A) <=75	age > 75	Δ (p-value)	(B) gmale	gender female	Δ (p-value)	(C) sign shocks ≤ 0	of shocks shocks > 0	$\Delta (p-value)$
Δ Index of Health Problems	0.103 (0.066)	0.211^{**} (0.082)	0.287	0.187^{**} (0.078)	0.126^{*} (0.067)	0.525	0.076 (0.200)	0.171^{*} (0.093)	0.670
Δ Self-reported Health	0.100 (0.116)	$0.152 \\ (0.167)$	0.788	0.044 (0.140)	0.179 (0.128)	0.453	-0.054 (0.341)	0.030 (0.169)	0.823
Self-reported Change in Health	0.272^{**} (0.127)	$0.204 \\ (0.151)$	0.718	0.207 (0.144)	$\begin{array}{c} 0.452^{***} \\ (0.128) \end{array}$	0.178	0.165 (0.388)	$0.251 \\ (0.197)$	0.844
Δ Mental Health Index	0.252^{*} (0.145)	0.210 (0.193)	0.858	0.141 (0.163)	0.310^{*} (0.158)	0.430	0.136 (0.427)	0.317* (0.190)	0.694
Survival	$0.072 \\ (0.059)$	0.333^{**} (0.144)	0.079*	0.192^{*} (0.110)	0.116 (0.086)	0.560	1.022^{***} (0.364)	0.198 (0.128)	0.034^{**}
controls (interacted):	~	~		~	~		~	~	
past wave's stock fraction	>				~		>		
time dummies	>				~		,		
other controls	>	~			Ń		~	/	
Coefficients on predicted wealth shock the difference between the two coeffici regressions; further comments as in Ta	as $((\widehat{\Delta W}_t/W$ ents; all con ble 5.	(t_{t-1}) , interaction trols are interaction.	cted with the two eracted with the su	respective sub ıbgroups; all v	groups, are vaves include	displayed; '∆ (p-val :d; dependent varial	lue)' indicates th bles are transforn	e significance] aed as in the b	tevel of aseline

Table 12: Subsample heterogeneity

	baseline	alternati	ve rescaling	using reported st	ock wealth changes
Dep. Var.	$\frac{\Delta S \& P500}{S \& P500} * s_{t-1} / W_{t-1}$	$\frac{\Delta S \& P500}{S \& P500} *_{St-1}/A_{t-1}$	$rac{\Delta S \& P500}{S \& P500} *_{s_{t-1}}/10,000$	$\frac{(4)}{(s_t - s_{t-1})/W_{t-1}}$	(5) $(s_t - s_{t-1})/10,000$
Health Problems Index	0.149^{***}	0.097^{***}	0.001	0.003	-0.000005
	(0.053)	(0.034)	(0.002)	(0.007)	(0.000033)
n	36,502	33,434	36,502	36,502	36,502
SRH	0.121	0.002	0.005	0.032^{**}	0.00092
	(0.100)	(0.063)	(0.003)	(0.015)	(0.000068)
п	46,220	42,011	46,220	46,220	46,220
$SR\Delta H$	0.338^{***}	0.150^{**}	0.012^{***}	0.010	0.00000
	(0.101)	(0.067)	(0.004)	(0.015)	(0.000050)
n	46,244	42,036	46,244	46,244	46,244
Mental Health	0.233^{*}	0.095	0.005	-0.005	0.000010
	(0.120)	(0.082)	(0.006)	(0.018)	(0.000073)
n	38,932	35,831	38,932	38,932	38,932
Survival	0.154^{**}	0.074^{*}	0.002	-0.014	-0.000032
	(0.073)	(0.045)	(0.003)	(0.016)	(0.000049)
n	36,804	33,504	36,804	36,804	36,804
controls:					
past wave's stock holdings	>	>	>	>	
time dummies	>	>	>	>	>
other controls	>	>	>	>	>
In column 3 and 4 the poorest c is set to one, thus constructing	quartile is excluded; in 5 and life time wealth as the sum e	6 retirees without stocks of current wealth and just	in (t-1) are excluded; in 7 and one annual income. Further	d 8 the life-time wealth comments as in Table	ı sum (see Data section) 5.

Table 13: Specification of Wealth Shocks

	(1) baseline	(2)		
	$\frac{\Delta S\&P500}{S\&P500} * \frac{s_{t-1}}{W_{t-1}}$	$\frac{\Delta S\&P500}{S\&P500} * \frac{s_{t-1}}{A_{t-1}}$		
Dep. Var.	retiree HH's	retiree HH's	non-retiree HH's	
Health Problems Index	0.166***	0.107**	0.002	
	(0.060)	(0.042)	(0.015)	
n	33,870	26,328	13,137	
SRH	0.211*	0.042	0.061**	
	(0.122)	(0.088)	(0.026)	
n	40,827	31,563	14,652	
$SR\Delta H$	0.215^{*}	0.122	0.017	
	(0.114)	(0.089)	(0.025)	
n	40,849	31,581	14,658	
Mental Health	0.312**	0.131	0.032	
	(0.144)	(0.108)	(0.026)	
n	36,550	29,073	13,944	
Survival	0.204**	0.137**	0.003	
	(0.090)	(0.067)	(0.006)	
n	31,710	24,478	11,327	
controls				
past wave's stock fraction	\checkmark	\checkmark	\checkmark	
time dummies	\checkmark	\checkmark	\checkmark	
other controls	\checkmark	\checkmark	\checkmark	
restricted to year>1999	\checkmark	\checkmark	\checkmark	

Table 14: Retirees vs. Non-retirees

Retiree HH's are definded as singles (or couples) that are (both) retired or out of the labor foce. For non-retiree HH's at least one HH member is employed or unemployed. A positive coefficient refers to a health improvement in the respective dependent variable; all dependent variable (including 'Survival') are standardized as explained in the data section; $\Delta X = X_t - X_{t-1}$; 'Surival' is 0 if respondent deceases before (t+1) and 1 otherwise; the estimation method used is OLS ('probit-adapted' OLS); standard errors, in parethesis, are clustered by individual; 'other controls' are: age, age², years of education, dummies for gender, race, initial marital status, degree and # months since previous interview.